Author's response to reviews

Title: Child-Pugh classification dependent alterations in serum leptin levels among cirrhotic patients: a case controlled study

Authors:

Fusun F Bolukbas (fusunbol@yahoo.ca)
Cengiz Bolukbas (drcengizbolukbas@hotmail.com)
Mehmet Horoz (mehmethoroz@yahoo.com)
Mahmut Gumus (mlgumus@ixir.com)
Mehmet Erdogan (erdogansena@ekolay.net)
Fadile Y Zeyrek (fadilezeyrek@hotmail.com)
Ali Yayla (dryayla@hotmail.com)
Oya Ovunc (oyaovunc@ekolay.net)

Version: 2 Date: 23 September 2004

Author's response to reviews:

September 23, 2004

Dear Editor,

Thanks for your attention that you have paid to our manuscript (MS: 1966270861342851) entitled as "Child-Pugh classification dependent alterations in serum Leptin levels among Cirrhotic patients: a case controlled study". Please find attached manuscript that revised in the light of reviewer comments and changed format as requested. The changes were provided with a point-by-point description at below:

1- Removing the figure title and number from the image file.

2- Figures were cropped as much as you requested and 'figure 1.a-b-c name'has been changed to Figure1, figure2 and figure3.

3- Removing all unnecessary capitalisations from the title, both in the text and the submission system.

4- Removing the shirt title from the title page

5- Deleting keywords section currently in the manuscript.

6- Renaming introduction section of the manuscript as 'Background'.

7- Removing all unnecessary capitalisations from all headings and the sub-headings in the manuscript. Italics were changed.

8- Including Competing Interests section between between the List of Abbreviations and Authors' contributions.

9- Providing the reference list in the journal style.

10- Correcting the minor grammatical errors.

12- In response to reviewer first comment, we deleted Leptin data in Fig. 1 and this data reported in the text.

13- In response to reviewer second comment, we pointed-out that among independent variables the viral etiologic factors were virus B and virus C, and the Child-Pugh's classification was intended A-B-C in the linear regression analysis.
Corresponding Author:
F. Fusun Bolukbas

Yenisehir mah. 4. sok. Altiparmak apt. Kat.2 Daire 3
63100 Yenisehir/ Sanliurfa TURKEY
Phone:+90 414 3155250
Fax: +90 414 3151181
e-mail: fusunbol@yahoo.ca